Author’s response to reviews

Title: Impact of a Postcrash First Aid Educational Program on Knowledge, Perceived Skills Confidence, and Skills Utilization Among Traffic Police Officers: A Single-Arm Before-After Intervention Study

Authors:

Menti Lastone Ndile (mentlast@yahoo.co.uk)
Gift Lukumay (giftgadiel@yahoo.com)
Karin Bolenius (karin.bolenius@umu.se)
Anne Outwater (anneoutwater@yahoo.com)
Britt-Inger Saveman (britt-inger.saveman@umu.se)
Susann Backteman-Erlanson (susann.backteman@umu.se)

Version: 3 Date: 09 Mar 2020

Author’s response to reviews:

In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
• Thank you for the comment. The role of funding agency has been described in the referred section with the sentence “The funding agency was not involved in the design of the study, data collection, data analysis, interpretation of data and writing the manuscript”

Many thanks for confirming in the section 'Ethics Approval and Consent to Participate', that the informed consent obtained was verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
• Thank you for the asking. Clarification has been made in the section to read “Before interviews were conducted, participants were verbally asked for their consent to participate in the study before administering written informed consent.